Mauna Loa Sol ar Observatory Cbserver’'s Log
Tue Apr 11 17:05:48 GVI 1995
Year: 95 Doy: 101
oserver: koon consol e Apr 11 17:01

WEATHER COMMVENT: Tue Apr 11 17:06:09 GMIr 1995
Thi ck al tocumul us overcast with cirrus above that, w nd=10 nph fromthe SW
tenmp=35 F.

LONL COWENT: Tue Apr 11 18:48:09 GMI 1995
Rermoved L00216 and installed L0O0218 in drive #0

WEATHER COMMENT: Tue Apr 11 18:48:42 GMI 1995
Clouds aren’t clearing, no observing today.

MKI'TT COMMENT: Tue Apr 11 21:20:35 GMTI 1995
Renoved O1, cleaned and replaced. Tightened dust cover nmounts. Exani ned
exi sting manual dust cover mechanismfor ways to notorize it.

DPMON COMVENT: Tue Apr 11 21:22:44 GVII 1995
Exami ned exi sting nonfunctional dust cover nechanismfor ways to get it working again

COMMVENT: Tue Apr 11 21:25:10 GMI 1995

Exam ned Nitrogen bottles and hydraulic lift (in done) to see how | can
attach the bottles to the Iift for use in blowi ng dust from Ols, bottles
have to be easily renovable for use in lens cleaning down in the trailer
and for refilling.

COMMENT: Tue Apr 11 21:28:30 GMI' 1995
TAPES:

*kkkk*k

LONML: LO0217 in drive #1

Tue Apr 11 21:29:28 GMI 1995: Fil emark



